Author's response to reviews

**Title:** Relationship between time-integrated disease activity estimated by DAS28-CRP and radiographic progression of anatomical damage in patients with early rheumatoid arthritis

**Authors:**

- Fausto Salaffi (fsalaffi@tin.it)
- Marina Carotti (marina.carotti@gmail.com)
- Alessandro Ciapetti (ciapetti.a@libero.it)
- Stefania Gasparini (gasparinistefania@libero.it)
- Emilio Filippucci (emilio_filippucci@yahoo.it)
- Walter Grassi (grassi.walter@gmail.com)

**Version:** 6  **Date:** 26 May 2011

**Author's response to reviews:** see over
Jesi, 26th of May 2011

Dear Sir,

Please find the revised version of the manuscript entitled “Relationship between time-integrated disease activity estimated by DAS28-CRP and radiographic progression of anatomical damage in patients with early rheumatoid arthritis”.

Major and minor revisions have been made as suggested. With regard to the link for the website address mentioned in the article we have verified that it works.

Best Regards

Fausto Salaffi
Marina Carotti
Alessandro Ciapetti
Stefania Gasparini
Emilio Filippucci
Walter Grassi